

onlinelibrary. wiley.com/doi/10.1029/2022JB025024 by Helmholtz-Zentrum Potsdam GFZ, Wiley Online Library on [22/12/2022]. See the Terms and Conditions (https://onlinelibrary

Wiley Online Library for rules

of use; OA

1699356

2022, 10, Dow

# JGR Solid Earth

# **RESEARCH ARTICLE**

10.1029/2022JB025024

### **Key Points:**

- Seismic velocities display a reduction trend by -0.45% preceding the 2018 Mt. Etna eruption
- Multi-parameter analysis shows a relationship between magma intrusion and the drop in seismic velocities
- Seismic velocity changes from ambient noise can be used in monitoring Mt. Etna for possible eruption precursors

#### Supporting Information:

Supporting Information may be found in the online version of this article.

#### Correspondence to:

P. Büyükakpınar, pinar.buyukakpinar@ingv.it; pinar@gfz-potsdam.de

#### Citation:

Büyükakpınar, P., Cannata, A., Cannavò, F., Carbone, D., De Plaen, R. S. M., Di Grazia, G., et al. (2022). Chronicle of processes leading to the 2018 eruption at Mt. Etna as inferred by seismic ambient noise along with geophysical and geochemical observables. *Journal* of Geophysical Research: Solid Earth, 127, e2022JB025024. https://doi. org/10.1029/2022JB025024

Received 21 JUN 2022 Accepted 3 OCT 2022

#### **Author Contributions:**

Conceptualization: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Raphael S. M. De Plaen, Thomas Lecocq Data curation: Pınar Büyükakpınar, Thomas King Formal analysis: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Daniele Carbone, Raphael S. M. De Plaen, Giuseppe Di Grazia, Thomas King, Marco Liuzzo, Giuseppe Salerno Methodology: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Raphael S. M. De Plaen, Thomas Lecocq

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Chronicle of Processes Leading to the 2018 Eruption at Mt. Etna As Inferred by Seismic Ambient Noise Along With Geophysical and Geochemical Observables

Pınar Büyükakpınar<sup>1,2,3</sup> , Andrea Cannata<sup>1,4</sup> , Flavio Cannavò<sup>1</sup> , Daniele Carbone<sup>1</sup> , Raphael S. M. De Plaen<sup>5,6</sup> , Giuseppe Di Grazia<sup>1</sup> , Thomas King<sup>1,7</sup> , Thomas Lecocq<sup>6</sup> , Marco Liuzzo<sup>8</sup> , and Giuseppe Salerno<sup>1</sup>

<sup>1</sup>Istituto Nazionale di Geofisica e Vulcanologia, Sezione di Catania, Osservatorio Etneo, Catania, Italy, <sup>2</sup>GFZ German Research Centre for Geosciences, Potsdam, Germany, <sup>3</sup>Institute of Geosciences, University of Potsdam, Potsdam, Germany, <sup>4</sup>Dipartimento di Scienze Biologiche, Geologiche e Ambientali, Università degli Studi di Catania, Catania, Italy, <sup>5</sup>Centro de Geociencias, Universidad Nacional Autónoma de México, Campus Juriquilla, Querétaro, Mexico, <sup>6</sup>Royal Observatory of Belgium, Brussels, Belgium, <sup>7</sup>National Buried Infrastructure Facility, University of Birmingham, Birmingham, UK, <sup>8</sup>Istituto Nazionale di Geofisica e Vulcanologia, Sezione di Palermo, Palermo, Italy

**Abstract** This work analyzes temporal variations of seismic velocities at Mt. Etna from August 2018 to February 2019. During this time period, a strong flank eruption accompanied by intense seismicity and ground deformation took place along a fracture that opened on 24 December 2018 at the base of the New South-East summit crater. Furthermore, two moderate earthquakes—the 6 October 2018  $M_L$  4.7 and 26 December 2018  $M_L$  4.8—associated with the volcanic activity were recorded. In this study, we computed cross-correlation functions (CCFs) between windows of seismic ambient noise to identify seismic velocity variations within the volcano edifice. We calculated daily CCFs at 16 stations for 120 interstation pairs using the vertical component in the 1.0–1.5 Hz frequency band. We observe that dv/v starts to decrease rapidly from the beginning of October 2018 and reaches approximately -0.45% in the pre-eruption period. The spatio-temporal distribution of seismic velocities shows that the reduction of dv/v mostly occurs in the vicinity of the summit and close to the flank area and is interpreted to be affected by magmatic intrusion at 0–3 km depth. To infer the source mechanism of this eruption, we compared these observations with volcano-tectonic earthquakes, volcanic tremor, volcanic degassing, gravity, and ground deformation data. Our study suggests that a relationship between magma intrusion and associated crack opening is responsible for the decrease of dv/v.

**Plain Language Summary** Seismic ambient noise offers a powerful method to monitor variations of seismic wave velocities due to volcanic and seismic activities. In this study, we investigated the magma intrusion before the 24 December 2018 Mt. Etna flank eruption. Due to the large number of permanent seismic stations in our study region, we retrieved a large data set to calculate the seismic velocity changes between August 2018 and February 2019. An unprecedented seismic swarm in the instrumental period occurred at the same time as the aforementioned volcanic activity. We found a drastic reduction in the seismic velocities in the pre-eruption period. We inferred this decrease is due to the opening of cracks caused by the intrusion of magma following significant stress changes in the crust. For the first time, we compared the seismic ambient noise analysis results with geophysical and gas geochemical parameters in the study region. The multi-comparison of different parameters also supports our results. It is proposed that the seismic ambient noise analysis will be implemented in real-time to monitor and detect possible precursor changes associated with the uprise of magma batches at Mt. Etna.

# 1. Introduction

Mt. Etna—the tallest and most active stratovolcano in Europe—is one of the best monitored volcanoes around the world since it is a natural laboratory for many disciplines (e.g., Bonaccorso et al., 2005; Patanè et al., 2013). On 24 December 2018, a flank eruption accompanied by an intense seismic activity took place and produced a dense rock equivalent (DRE) lava volume of about 2 Mm<sup>3</sup> (Figure 1; Laiolo et al., 2019). This volcanic crisis high-lighted the importance of early warning systems to mitigate the potential damage in the surrounding settlements of Mt. Etna. Despite the significant improvements of technological progress and research, there still exists an uncertainty in forecast volcanic eruptions even with multi-parametric observations (e.g., Loughlin et al., 2015).



Project Administration: Flavio Cannavò Software: Raphael S. M. De Plaen, Thomas Lecocq Supervision: Andrea Cannata, Flavio Cannavò

Validation: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Daniele Carbone, Raphael S. M. De Plaen, Giuseppe Di Grazia, Thomas King, Thomas Lecocq, Marco Liuzzo, Giuseppe Salerno

Visualization: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Raphael S. M. De Plaen, Thomas Lecocq Writing – original draft: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Daniele Carbone, Raphael S. M. De Plaen, Giuseppe Di Grazia, Thomas Lecocq, Marco Liuzzo, Giuseppe Salerno

Writing – review & editing: Pınar Büyükakpınar, Andrea Cannata, Flavio Cannavò, Raphael S. M. De Plaen Recent advances show that the measurement of seismic velocity by passive interferometry is sensitive to the cycles of magma pressurization and the redistribution of melt within the plumbing system (Brenguier et al., 2008; Duputel et al., 2009). Hence, the seismic ambient noise analysis helps shed a new light on the other observables to better understand the cycles of magma recharge before an eruption.

Seismic velocity changes have been observed at the same time as volcanic eruptions around the world due to magma intrusion and intense pressurization—for example, at Kīlauea (Donaldson et al., 2017; Olivier et al., 2019), Piton de la Fournaise (Clarke et al., 2013; De Plaen et al., 2016), Mt. Þorbjörn-Svartsengi (Cubuk-Sabuncu et al., 2021), and Sierra Negra volcano (Ruiz et al., 2022). To detect the variations of seismic velocities in the order of <1%, seismic ambient noise and cross-correlation functions (CCFs) are widely used (e.g., Brenguier et al., 2008; Donaldson et al., 2017). Recently, early warning systems have been implemented to follow the seismic velocity changes in real-time (e.g., Hutapea et al., 2020).

Seismic noise interferometry was previously studied at Mt. Etna by Cannata et al. (2017) during 2007–2015. They found variations in seismic velocities due to the medium seasonal changes in the range -0.8% to 0.8% for the frequency band of 0.1-0.3 Hz. Furthermore, De Plaen et al. (2019) analyzed the 2013–2014 eruptive period and revealed seismic velocity changes between 0.3% and -0.2% in the summit stations indicating the intense pressurization accompanying paroxysmal eruptions. Moreover, a 2% drop of seismic wave velocity in the north-eastern flank of Mt. Etna during the 2002–2003 eruption was observed in the study of Cannata (2012) by using the coda wave interferometry applied on repeating volcano-tectonic earthquakes (VTs).

In this work, we apply the seismic ambient noise method to Mt. Etna in the period from August 2018 to February 2019 during which the aforementioned strong flank eruption occurred on 24 December 2018, accompanied by the most intense swarm of VTs recorded during the last 20 years (Figure 1 and Figure S1 in Supporting Information S1). To infer the source mechanism of this variation, we integrated seismic ambient noise results with those of VTs, focal mechanisms, volcanic tremor, gas geochemistry, gravity, and ground deformation data recorded by the multi-parametric monitoring network of Istituto Nazionale di Geofisica e Vulcanologia (INGV; Figure 1a).

#### 1.1. Volcanological Framework

During the last decade, Mt. Etna has been characterized by frequent eruptive activities, involving most of the summit craters along with lateral fissures (Figure 1a). In particular, from 2011 to 2013, the New South-East Crater (NSEC) produced more than 40 explosive eruptions culminating in lava fountaining (e.g., Behncke et al., 2014; De Beni et al., 2015). From mid-December 2013 to the end of 2014, the lava fountains at NSEC were replaced by violent Strombolian activities mainly at NSEC (e.g., De Beni et al., 2015) as well as by an eruption from a fissure opened at 3,000 m elevation on the lower eastern flank of the North-East Crater (NEC) during July–August 2014 (e.g., Spina et al., 2017; Viccaro et al., 2016). On 28 December 2014, another lava fountain was produced by NSEC (Gambino et al., 2016). In December 2015 and July 2016, violent explosive activities affected also Voragine Crater (VOR; Cannata et al., 2018). In 2017, the activity became again more effusive with a series of weak Strombolian episodes and lava flow emissions between February and April (Viccaro et al., 2019).

During the first months of 2018, the eruptive activity of Mt. Etna was characterized at NSEC and NEC by mild ash emissions that intensified turning into a Strombolian activity in July. In August, the Strombolian activity involved the Bocca Nuova (BN), NEC, and NSEC, and on 23 August about at 18:00 UTC, the last crater started an effusive activity that gave rise to modest lava flows on the Eastern flank of the NSEC cone (Crafford & Venzke, 2018). The Strombolian activity from BN, NEC, and NSEC and the discontinuous effusive activity from NSEC carried on up to December (Borzi et al., 2020). On 24 December at 11:11 UTC, an eruptive fissure opened at the eastern base of NSEC and propagated in the direction SSE (Figure 1a). This eruption, characterized by both explosive and effusive activities accompanied by significant ground deformation and seismic activity, ended on 27–28 December (e.g., Bonforte et al., 2019; Cannavo' et al., 2019; Carbone et al., 2019; Chauhan et al., 2020). This eruptive episode was also accompanied by significant variations in  $CO_2$  and  $SO_2$  fluxes and  ${}^{3}He/{}^{4}He$  ratios (e.g., Gurrieri et al., 2021; Paonita et al., 2021). The following weeks were characterized by mild ash emissions from the summit craters.



# Journal of Geophysical Research: Solid Earth



Figure 1.



#### 1.2. Seismological Settings

Mt. Etna is located at the front of the Apennine–Maghrebian Chain on an Early–Middle Pleistocene foredeep clayey succession (e.g., Branca et al., 2011; Monaco et al., 2005). The complex interactions between volcano dynamics and tectonics gave rise to several active faults, causing moderate earthquakes with  $M_L < 5.0$  (Figure 1b; Alparone et al., 2015, 2020). The well-known faults structures in the region associated with the 2018 eruption are (Figure 1b; De Novellis et al., 2019): (a) Pernicana Fault (PF) lies on the volcano's north-eastern flank with an oblique left-lateral strike-slip mechanism (e.g., Azzaro et al., 2012; Bonforte et al., 2007); (b) Ragalna Fault (RF), mainly characterized by a right-lateral strike-slip mechanism, is located on the southwest flank (e.g., Azzaro et al., 2012; Rust & Neri, 1996); (c) Fiandaca Fault (FF) is a part of the main Timpe system in the southeast and shows the dextral oblique-slip behavior similar to RF (e.g., Azzaro et al., 2012; Villani et al., 2020).

The Mt. Etna eruption in 2018 was accompanied by especially the intense seismic activity (Figure 1b; Alparone et al., 2020). The swarm of VT earthquakes could give clues about the magmatic intrusions. The two largest earthquakes that took place during the analyzed time interval occurred on 6 October 2018 with  $M_1$  4.7 and on 26 December 2018 with  $M_1$  4.8. The former occurred in the south-western flank close to the RF at 6 km depth. The latter was located in the south-eastern flank at a very shallow depth (1.2 km) along the FF (Figure 1c) and caused a significant damage along the surface rupture (Civico et al., 2019; Villani et al., 2020). To study the time-space evolution of the seismicity and its relationship with the seismic ambient noise results, we used the seismic events relocated by Alparone et al. (2020) by double-difference (Zhang et al., 2009) and the 3D velocity model by Alparone et al. (2012). This catalog contains approximately 1,700 earthquakes that took place during the interval 6 October to 26 December 2018 (Figures 1b and 1c, and Figure S1 in Supporting Information S1). Most of the seismicity occurred in a very short time interval, accompanying the 24 December intrusion and clustering in two rock volumes along the eruptive fracture with epicenters aligned in a N-S direction and slightly southeast of the summit (Figure 1b; Alparone et al., 2020). It is also worth noting that earthquakes predominantly occurred in the upper ~6 km b.s.l. of the Mt. Etna (Figure 1c and Figure S1 in Supporting Information S1). For a better visualization of the seismicity in the region for that period, a 3D animation can be found in Movie S1 in Supporting Information S1.

#### 2. Materials and Methods

#### 2.1. Seismic Data and Processing

#### 2.1.1. Seismic Ambient Noise

In this study, we used continuous data from 16 stations equipped with broadband three-component Trillium seismometers (40s cut-off period, Nanometrics<sup>TM</sup>), operated by INGV-Osservatorio Etneo (OE; Figure 1a and Figure S2 in Supporting Information S1). We processed their vertical component using the MSNoise package (Lecocq et al., 2014). The raw data were band-pass filtered between 0.01 and 8.0 Hz, down-sampled from 100 to 20 Hz, and de-trended. Daily traces were segmented into 1-hr windows, winsorizing (clipping) at 3 Root Mean Square (RMS), and spectral whitening in the 1.0–1.5 Hz frequency band were applied before cross-correlation (Bensen et al., 2007; Lecocq et al., 2014). This frequency band was chosen after a sensitivity analysis, which showed that this band gives the most stable results in terms of both dv/v time series and sharper variations focused on the main eruptive events. In the analysis, the variations of dv/v were computed for other four frequency bands: 1.0–2.0, 1.2–1.6, 1.2–1.4, and 1.4–1.6 Hz (Figure S3 in Supporting Information S1). Seismic velocity changes were also calculated for the lower frequency bands (0.1–1.0 and 0.5–1.0 Hz) and higher frequency bands (1.6–1.8, 1.6–2.0, 1.8–2.0, and 2.0–2.5 Hz) and further significant variations were not found. Hourly windows were then stacked

**Figure 1.** (a) The station distribution in the study area. Triangles show the location of the broadband seismic stations. Black squares show the GPS stations (EPLU, EPDN, ECPN, and EINT). The diamond presents the gravimeter (SLN). Reverse triangles point the EtnaGas network to measure  $CO_2$  flux. Octagons indicate the sites of the Mt. Etna Flame-DOAS network. The contour lines are plotted for every 1,000 m of elevation. The yellow box points to the location of the inset figure. The inset figure shows the digital elevation model of the summit of Mt. Etna (VOR, Voragine; BN, Bocca Nuova; NEC, North-East Crater; SEC, South-East Crater; NSEC, New South-East Crater) from Neri et al. (2017). The light-green line shows the eruptive fissures. (b) The overview map of the volcano-tectonic earthquakes (VTs). The stars indicate the epicenters of moderate earthquakes M > 4.0 that accompany the 2018 eruption at Mt. Etna. The colored dots show the seismic events from 2018 January until 2019 April in a time scale (Alparone et al., 2020). Brown lines show the fault lines. PF: Pernicana Fault, RF: Ragalna Fault, FF: Fiandaca Fault. The focal mechanisms are compiled from different sources as follows: orange, yellow, blue, green, and gray beach balls are taken from GCMT, Pondrelli (2002), Scarfi et al. (2013), Scognamiglio et al. (2006), and GEOFON, respectively. The black mechanisms were calculated in this study. (c) The depth cross-section of the VTs with respect to longitude.



for each day and cross-correlation quality and stability were analyzed in both the time and frequency domains by visual check (Figure S4 in Supporting Information S1). The resulting daily CCFs were then stacked using the time-frequency phase-weighted stacking method (tf-PWS; Schimmel & Paulssen, 1997) for 5, 10, and 30 days for each station pair to increase the signal-to-noise-ratio.

Daily variations in seismic velocity (Figure 2a) were computed using the Moving Window Cross-Spectrum analysis (MWCS; Ratdomopurbo & Poupinet, 1995) between a daily and a reference CCF, which comprises the January 2018 to July 2018 time period in our study and corresponds to the pre-eruption stage. Unlike the stretching technique of Sens-Schönfelder and Wegler (2006), the MWCS method only uses phase spectra to estimate seismic velocity changes, which reduces the contamination due to temporal variations in the source of seismic noise on the seismic velocity changes (Zhan et al., 2013). We compute time shifts, dt, in the coda of daily CCFs relative to the reference CCF. Time shifts and coherency c between the reference and daily CCFs are calculated after the 1 km/s arrival in the coda in 30 s windows and the dt measurements are kept when the coherency c is equal or higher than 0.5. In the sensitivity analysis, the coherency c was changed from 0.5 to 0.8 and any notable alteration for the further computations was not observed (Figure S5 in Supporting Information S1). The regressing time shifts dt from each window in the causal and acausal part of the coda then provided the daily time shift dt/t. Assuming a homogeneous change throughout the sampling medium, the relative difference in travel time dt is related to the change in the seismic velocity dv as -dt/t = dv/v. The resulting time series for each possible pair of stations on Etna volcano were finally combined in a single median variation of seismic velocity for the network (Figure 2a). Furthermore, to map the "epicentral area" showing the strongest change in seismic velocity, the method of Brenguier et al. (2014) is applied. According to this approach, a velocity change value, computed as the mean of the velocity changes measured for the pairs with which a station was involved, is assigned to each station (Figure 3). To estimate the depth sensitivity of the analysis, we calculate the partial derivatives of the fundamental-mode Rayleigh-wave phase velocities (Herrmann, 2013). The sensitivity kernels show that dv/vmeasurements in our study indicate velocity variations approximately within the upper 3 km of the crust (Figure S6 in Supporting Information S1) considering the velocity model of Aloisi et al. (2002).

#### 2.1.2. Volcano-tectonic Earthquakes

We obtained moment tensors (MTs) for the events  $M_L > 3.9$  based on a Bayesian bootstrap probabilistic joint inversion method by using the Grond software (Heimann et al., 2018; Kühn et al., 2020). The method provides the uncertainties for the model parameters of resulting moment tensor inversion (e.g., depth) and has been applied to different regions recently—for example, in Southeastern Anatolia (Büyükakpınar et al., 2021), Zagros Simply Folded Belt (Jamalreyhani et al., 2021), and AlpArray seismic network (Petersen et al., 2021). The MTs were calculated via waveform inversion in both time and frequency domains using three-component data (R, T, and Z). Synthetic seismograms were computed using pre-calculated Green's functions (Heimann et al., 2019) based on the combination of two velocity models for the crust (Aloisi et al., 2002) and upper mantle (AK135, Kennett et al., 1995). We used the frequency band of 0.03–0.08 and 0.02–0.05 Hz depending on the source-to-receiver distance for events smaller than  $M_L$  4.3 and the events larger than  $M_L$  4.3, respectively. We retrieved the MTs in full waveform inversion for the two largest earthquakes ( $M_L = 4.7$  and 4.8) and the double-couple solutions for the smaller earthquakes (Figure 1b and Table S1 in Supporting Information).

#### 2.1.3. Volcanic Tremor

Volcanic tremor is one of the most important and common precursory phenomena of eruptive activity because of its strict relationship with fluid flows through the volcano's feeding system (Chouet & Matoza, 2013). This is particularly true at Mt. Etna where important variations of this seismo-volcanic signal have been observed, both in the medium and short terms, before the onset of eruptions (Cannata et al., 2018; Cannavo' et al., 2019). Volcanic tremor is commonly addressed to as RMS amplitude of the seismic signal measured over a certain time window and frequency band. In this work, we calculated the daily average of the RMS amplitude of seismic signal in velocity (m/s), recorded at a seismic station, ECPN, located at about 1 km from the summit craters (Figure 1a). We chose to filter the seismic signal in the frequency band 0.5–2.5 Hz as it contains most of the energy of volcanic tremor at Mt. Etna during upcoming unrest (Figure 2b and Figure S7 in Supporting Information S1).



# Journal of Geophysical Research: Solid Earth

10.1029/2022JB025024



**Figure 2.** (a) Average velocity variation of 5-day moving window (dv/v) for all station pairs with its uncertainty (gray error bars). (b) Root Mean Square amplitude of volcanic tremor in velocity (m/s) filtered in the frequency band 0.5–2.5 Hz. (c) Gravity signal observed at SLN station (iGrav#16). (d) Normalized CO<sub>2</sub> flux from the volcano flanks, measured by the EtnaGAS network. (e) The SO<sub>2</sub> flux measured via FLAME network. (f) The areal strain of the summit area calculated using the EPLU, EPDN, ECPN, and EINT GNSS stations. The vertical dashed red lines show the 23 August Strombolian and 24 December eruption. The dashed green lines indicate the 6 October 2018  $M_L$  4.7 and 26 December 2018  $M_L$  4.8 earthquakes.





**Figure 3.** The spatio-temporal variations of dv/v and the broadband seismic stations, shown by red triangles and used in the seismic ambient noise analysis. The three columns show three different time ranges for phase 1, phase 2, and phase 3. The epicentral locations of volcano-tectonic earthquakes are shown by gray circles for each phase. The first, second, and third rows correspond to 5, 10, and 30 days moving windows, respectively. The contour lines show 1,000 m elevation interval. The light-green line shows the eruptive fissures in phase 3. The interpolation was applied by the triangulation method using the GMT software (Wessel et al., 2019).

#### 2.2. Gravity Data

The study of temporal gravity changes can provide unique insight into the processes driving volcanic activity, since gravity is the only parameter that is directly influenced by the redistribution of underground mass (Carbone et al., 2017). At Mt. Etna, a mini-array of continuously running superconducting gravimeters (SGs) has been in operation since 2016 (Carbone et al., 2019). These instruments provide better quality data than the more widely used spring gravimeters, since they are not affected by environmental parameters (Van Camp et al., 2017). Indeed, thanks to their high precision and long-term stability, SGs allow to track gravity changes in the order of



 $1-2 \mu$ Gal, occurring over time scales between minutes and several months. In this study, we utilize the gravity time series from iGrav#16 at SLN (Figure 1a), jointly with the other available time series (Figure 2c).

#### 2.3. Soil CO<sub>2</sub> and SO<sub>2</sub> Flux Data

The geochemical observations from  $CO_2$  and  $SO_2$  flux data play a significant role in understanding the relations between degassing and volcanic/tectonic activities. We used the  $CO_2$  data from ETNAGAS network consisting of 14 stations operated by INGV-Palermo (Figure 1a). To remove the effects of temperature, pressure, and wind variations on the geochemical parameters, specific filters were applied to the data set (Figure 2d; Gurrieri et al., 2021; Liuzzo et al., 2013; Paonita et al., 2021). Furthermore,  $SO_2$  flux data from UV-scanner Flame-DOAS network from nine stations established by INGV-OE (Figure 1a) were used in this study (Figure 2e; Paonita et al., 2021; Salerno et al., 2009).

#### 2.4. Ground Deformation

We used the GNSS data to evaluate the ground deformation in the considered period. The GNSS permanent network run by INGV-OE consists of 39 stations mainly equipped with Leica Geosystems instruments (Figure 1a). The GNSS raw data coming from the stations are processed on a daily basis using the GAMIT/GLOBK software (Herring et al., 2010). The processed GNSS final daily solutions are referred to the geodetic reference frame specifically designed for the area (Palano et al., 2010). To characterize the volcano deformation state, we calculated the areal strain in the volcano summit area by using the GNSS daily solutions (Figure 2f).

#### 2.5. Volumetric Strain Change

Volumetric strain models can provide insights into plumbing system dynamics (Donaldson et al., 2017) as well as can help model seismic velocity changes. For example, seismic velocities can decrease or increase during magma uprising due to the opening and closing of cracks, respectively (Bennington et al., 2015; Brenguier et al., 2008; Hotovec-Ellis et al., 2015; Obermann et al., 2013). The inflation of a pressure source causes extensional and compressional volumetric strain above and at the sides of the source, correspondingly. Deeper sources host larger extension and vice versa (Donaldson et al., 2017). We calculated the volumetric strain due to the ellipsoidal sources proposed in Mattia et al. (2020) for the April–October and October–December pre-eruption periods, placed at ~5 km b.s.l. and with volume variations of 12 and 21  $10^6$  m<sup>3</sup>, respectively. We calculated the volumetric strain out to 10 km from the source by computing the infinitesimal strain tensor from the Yang analytical solutions within the half-space (Yang et al., 1988) (Figure S8 in Supporting Information S1).

#### 3. Results

The multidisciplinary observation of the period surrounding the December 2018 eruption of Mt. Etna volcano yielded distinct time series, which, taken together, helps unfold the events leading to the eruption (Figure 2). We jointly analyzed all the data sets and divided the period preceding and accompanying the 2018 eruption into three distinct phases as chronologically as follows (Figure 2): (a) 1 August to 5 October, (b) 6 October to 31 October, and (c) 1 November to 27 December. These phases were determined according to the major events, such as (a) the strongest seismic event preceding the eruption, the 6 October 2018  $M_L$  4.7 earthquake; (b) the variation in the temporal trend of the gravity signal, 31 October 2018; and (c) the end of the eruption period as well as the increase of the dv/v, 27 December 2018, respectively.

Seismic velocities in Figure 2a do not display any particular perturbation during phase 1. A sharp drop of about 0.45% of dv/v at the beginning of October continuing the reduction till the 24 December eruption period is quite evident in the frequency band of 1.0–1.5 Hz. Furthermore, similar reduction trends of dv/v are also observed in other frequency ranges: 1.0–2.0, 1.2–1.6, 1.2–1.4, and 1.4–1.6 Hz (Figure S3 in Supporting Information S1). After the eruption onset, dv/v started to increase recovering the whole variation accumulated in October, settling again on values close to the ones observed prior to the pre-eruptive period. The normalized dv/v histogram plot showing the variation of dv/v in time for all the station pairs can be found in Figure S9 in Supporting Information S1.

The VTs  $M_L > 3.9$  display mostly the strike-slip mechanism in the vicinity of Mt. Etna during the seismic swarm that took place in 2018 (Figure 1). The source parameters (strike, dip, and rake) that we calculated are in good agreement with the previous reports given by different sources (e.g., GCMT, Pondrelli, 2002; Scarfi et al., 2013; Scognamiglio et al., 2006). However, we estimated shallower centroid depths <4 km for most of the earthquakes. The two largest earthquakes, 6 October 2018  $M_L$  4.7 and 26 December 2018  $M_L$  4.8 indicate 8.7 ± 1.1 and 3.8 ± 0.7 km centroid depths, respectively. An example of waveform fits for the former earthquake is illustrated in Figure S10 in Supporting Information S1. The centroid depths that we obtained are compatible with the hyportiang Information S1.

As for the volcanic tremor, Figure 2b shows two clear increments preceding and accompanying both the Strombolian activity in August (phase 1) and the eruptive activity in December (phase 3). In particular, volcanic tremor showed a sharp amplitude increase preceding by 2 hr the onset of the activity that occurred on 23 August 2018. As for the latter, the volcanic tremor amplitude started increasing at about 08:30 on 24 December 2018 and then a couple of hours before the eruptive fissure opening. Furthermore, it is worth noting that, since the end of October, we observe a slow and steady increase in the average amplitude of the volcanic tremor until the beginning of December eruption. This could be due to the motion of magmatic masses toward more superficial portions of the volcano's plumbing system with consequent pressurization of the shallow part of the plumbing system as observed during previous unrest periods (Cannata et al., 2015, 2018). Conversely, a decreasing trend occurs after the end of the eruption, probably caused by the magmatic pressure loss.

The most striking feature in the gravity data from SLN (Figure 2c) is the increase phase, which occurred during October and early November (phase 2). Gravity increased by slightly more than 10  $\mu$ Gal with an average rate of 0.4  $\mu$ Gal/day. Since no significant elevation change took place at SLN, the free-air effect could be neglected and the resulting gravity change would be only associated with underground mass changes. It is important to note that an intense rainfall took place in mid-October 2018 (Figure S11 in Supporting Information S1), implying that a part of the above increase in the signal is likely due to changes in groundwater mass (Carbone et al., 2019). Since we do not have precise knowledge of the amplitude of the volcano-related gravity change, nor of the depth of the volcanic mass source, we cannot provide a reliable figure for the amount of mass involved in the redistribution process.

With regard to the variations of  $CO_2$  measured by the EtnaGAS network, the most intense oscillations can be observed between August and October 2018 (phases 1 and 2) with three increase-decrease variation cycles of  $CO_2$  flux, reaching the highest intensity in mid-October (Figure 2d). A further significant trend is the period of decrease, which begins in mid-October and continues steadily until December, reaching minimum values that coincide with the start of eruptive activity. Previous to this work, similar  $CO_2$  cycle variations during eruptive periods have been correlated to inputs of magmatic ascent from an intermediate plumbing system (Gurrieri et al., 2021; Liuzzo et al., 2013).

Throughout August 2018 and December 2018, the bulk daily mean plume  $SO_2$  flux (Figure 2e) released by the summit and the eruptive fracture of the 24 December 2018 eruption was featured by a main degassing stage with values higher of those recorded over non-eruptive quiescent conditions at Mt. Etna (~5,000 tons/day). Between November 2018 and January 2019 (phase 3), the  $SO_2$  mass emission rate increased up to mean values between 10,000 and 12,000 tons/day. Specifically, over the eruptive activity observed with the start of the activity at the NSEC crater. The  $SO_2$  emission rates reverse its trend since the second weeks of January to gradually decrease toward the typical degassing quiescent values of Mt. Etna.

The long lasting inflation period started in 2014–2015 and was shortly interrupted during the five main eruptive episodes that occurred till the December 2018 flank eruption. These episodes had not affected the inflating trend whose source was estimated to be located at a depth of about 7 km b.s.l (Cannavò, 2019). The trend was suddenly interrupted the day of eruption, when the deformation data showed a dike-like intrusion pattern, followed by a deflation period (Figure 2f).

Spatio-temporal seismic velocity changes and the corresponding seismicity were shown for 5, 10, and 30 days smoothed window in Figure 3 for different phases. The dv/v does not show any considerable variation for phase 1 although a Strombolian activity was recorded on 23 August at the summit craters (Figures 3a, 3d, and 3g). The





**Figure 4.** Conceptual model of the different phases of the magmatic feeding system preceding the December 2018 eruption. The diagrams illustrate W-E slices of Mt. Etna volcano down to ~ 10 km b.s.l. (a) Phase 1, deep recharge of the reservoir at a depth of 7–12 km b.s.l. A small Strombolian eruption on 23 August 2018, probably fed by a shallow source. (b) Phase 2, magma transfer toward the intermediate reservoir increasing the stress and the surrounding seismicity, leading to the ML 4.7 earthquake as well as to the reduction of seismic velocities. Rising magma filled the cracks in the system causing an increase of gravity and release of CO<sub>2</sub>. (c) Phase 3, 24 December 2018 flank eruption through two dikes until the shallow  $M_L$  4.8 earthquake. The flank sliding helps release pressure and abort the ongoing fissure eruption.

seismic velocities indicate a strong reduction in Phase 2 down to -0.45% mostly in the vicinity of the summit and south-eastern part of the summit. The number of VTs in this period increases sharply (Figure S1) and the hypocenters become shallower between 3 and 6 km (Figure 3b, 3e, and 3h; Alparone et al., 2020). The largest seismic event in this period, the 6 October 2018 M<sub>L</sub> 4.7, and its aftershocks are located in the southwest near RF. For Phase 3, the dv/v remains at the same level as at the end of Phase 2 (Figure 3c). A strong seismic swarm took place in the south-eastern flank, mostly shallow part of the crust, in the vicinity of the observed largest decrease of seismic velocity during this phase (Figure 3c, 3f, and 3i; Figure S1 in Supporting Information S1).

According to the strain analysis, a significant increase in both the areal and volumetric strains was observed in October–December (phases 2 and 3), thus agreeing with the observed decrease in dv/v. Moreover, the increase in the volumetric strain spatially coincides with the volume sensitive to the dv/v analysis (Figure S8 in Supporting Information S1). Indeed, most of the seismic stations showing large dv/v variations (Figure 3c) are located in correspondence with a near-surface extensional area above a 4-km thick bulk where volumetric strain changes of ~0.5 nanostrain are modeled.

#### 4. Discussion

The period preceding the 24–27 December eruption can be divided into three distinct phases based on the important changes in key observables (Figure 4). The boundary between the first and the second phases is the 6 October 2018  $M_L$  4.7 earthquake, coinciding with the onset of the decreasing trend of dv/v. The boundary between the second and the third phases is the trend change in the gravity signal, from increasing trends to stable values, on 31 October 2018. Finally, the end of the third phase coincides with the end of the eruption as well as with the apparent increase of the dv/v.

In the first phase, up to 6 October 2018, the relative stability of most observables does not indicate any clear precursors, even though this period includes the 23 August 2018 Strombolian eruption (Figure 4a), except for the  $CO_2$  that shows a long-term precursor since the beginning of 2017 (please see Paonita et al., 2021). The 23 August 2018 eruption was considered weak and its source shallow (Borzi et al., 2020). Its only measurable precursory activity was an increase in volcanic tremor amplitude, which began by 2 hr before the eruption. It is worth mentioning here that the volcano displayed a resting period for 15 months before this phase since after the

February–April 2017 eruption (Borzi et al., 2020). This phase was specifically associated with the deep recharge of the reservoir at 7–12 km b.s.l. as evidenced by a progressive increase of  $CO_2$  flux (Figure 2d), of the  $CO_2/SO_2$  ratio and of  ${}^{3}He/{}^{4}He$  ratios (Gurrieri et al., 2021; Paonita et al., 2021) in agreement with the pressure source estimated by deformation (Cannavo' et al., 2019). The seismicity in the corresponding period is scattered down to 30 km of depth (Figure 1c; Alparone et al., 2020).

During the second phase (6-31 October 2018), one of the most striking features is the seismic velocity decrease, which occurred in early October (Figure 2a). Similar short-term variations in dv/v before volcanic eruptions have been observed in many studies around the world. Brenguier et al. (2008) and Obermann et al. (2013) observed dv/v decreases preceding different eruptive periods at Piton de la Fournaise volcano. They show that seismic velocity reduction is spatially localized close to the magma reservoir and it is likely related to opening of the cracks in the volcano edifice, caused by magma pressurization. Studies at Kilauea volcano showed that seismic velocity increases and decreases stem from continual deflations and inflations of the summit, respectively (Donaldson et al., 2017). Moreover, Olivier et al. (2019) observed a velocity decrease 10 days prior to the 2018 eruption of the Kīlauea volcano. Recently, 17 days before the 2018 Sierra Negra volcanic eruption, a velocity drop has been revealed by Ruiz et al. (2022). At Mt. Etna, the dv/v reduction during that period of phase 2 would be likely due to sudden pressurization of the deep magma batch (Figure 4b), causing shallow (within 3 km) crustal variations, which are opening of fractures just above a source that is pressurizing before migrating (see also, Donaldson et al., 2017). The strain models for such a source (Figure S8 in Supporting Information S1) produce an extensional regime directly above and below the pressure source models. It is important to mention here that the dv/v could also be significantly impacted by variations of water storage in the top few hundred meters of the crust (e.g., Lecocq et al., 2017). However, the large rainfall only affected the region almost 10 days after (from mid-October) seismic velocity reduction started (Figure S11 in Supporting Information S1). Our study supports the idea that reduction of medium seismic velocity is due to crack openings within the extensional regime determined from the volumetric strain analysis as was proposed in several previous studies (Brenguier et al., 2008; Donaldson et al., 2017; Hotovec-Ellis et al., 2015; Obermann et al., 2013). Similar patterns were observed at Mt. Etna as well. De Plaen et al. (2019) suggested that seismic velocity decreases result from intense pressurization of magma within the plumbing system for 2013–2014 paroxysmal eruptions. Patanè et al. (2003) indicated that magma migration within the plumbing system at Mt. Etna led to the dilatation of the volcano edifice preceding the 2002 eruption.

During phase 2, an earthquake of magnitude 4.7, the largest event since 2009, occurred on 6 October 2018 along the RF (Figure 4b). Magma ingression could trigger seismicity in the proximity of ascending magma due to changes in physical properties (e.g., stress state, elastic moduli, density, and seismic velocity) and might have triggered this earthquake. Following that, the seismicity increased in the surrounding areas and became shallower  $(H \le 6 \text{ km b.s.l.}; \text{ Figure 1 and Figure S1 in Supporting Information S1})$ . Alparone et al. (2020) interpreted this seismicity increase as the first clear sign of unrest, which could be linked to the progressive ascent of magma through the upper crust. Due to the growing rate of fractures, the volcanic edifice is likely subjected to the accumulation of damage, which is supported by abundant shallow seismicity as an indication of magma intrusion in this period. Spatio-temporal distributions of dv/v in the vicinity of the summit area within 10 km well coincide with the seismicity distribution and the larger changes in velocities (Figure 3). The temporal dv/v drop in this phase coincides with the  $M_L$  4.7 earthquake. To better understand the causal relationship among magma ascent,  $M_1$  4.7 earthquake, and seismic velocity drop, we calculated the dynamic stress change (Hill & Prejean, 2015) associated with the  $M_1$  4.7 earthquake and it was equal to 0.15 MPa based on the maximum peak ground velocity observations at the used seismic stations (Figure 1). Such a dynamic stress change value seems to be too low to be responsible for the observed 0.45% (1.0–1.5 Hz) and 0.20% (0.1–1.0 Hz) dv/v drop (Figure S3) compared to previous studies (e.g., Brenguier et al., 2014; Taira & Brenguier, 2016). Hence, both the 6 October earthquake and the dv/v change are likely to be the effects of magma ascent. However, we cannot exclude that the 6 October earthquake played a role in intensifying the dv/v change.

Concurrently with dv/v drop in phase 2, there is an increase in the gravity signal by ~10 µGal. The dv/v decrease and the gravity increase are accompanied by an increase in  $SO_2$  and a renewed input of  $CO_2$ . When we consider the changes of all the mentioned signals in this period, phase 2 can be interpreted as follows. The deep increase of pressure, which led to the 6 October 2018 main earthquake, caused the opening of fractures and the ascent of magma simultaneously. This synchronous phenomenon generated the decrease of dv/v but not a decrease in gravity as fracturing processes and magma uprise compensate for each other in terms of mass. Subsequently, the



magma continued flowing upward and it gave rise to an increase in mass because of compression of pre-existing gas bubbles, gas dissolution (due to the pressure increase), and possibly filling of pre-existing pore space while the fracturing process ceased. Furthermore, GPS areal strain data (Figure 2f) show a steady increase that could indicate that the volcano inflating trend did not change even in this period.

In the last phase, 1 November to 26 December, volcanic tremor amplitude, GPS areal strain, and SO, flux increased while the gravity signal remained stable. The increase of the first two observables could be due to the pressurization of the upper plumbing system. The growth of SO<sub>2</sub> flux likely points to shallow magma batch degassing. The flattening of gravity signal might be linked to no mass variation and then no important magma migration/shallow intrusion (Figure 4c). In this phase, dv/v showed slight oscillations similar to the ones observed during the second part of phase 2. We interpret this phase as characterized by a stable shallow magma batch whose exsolved gases are increasing the measured SO<sub>2</sub> flux. The increase in the fluid dynamics within the shallow plumbing system can also explain the intensification of the volcanic tremor amplitude, whose source is generally associated with the dynamic interactions of fluids with surrounding rocks along magma pathways (Chouet & Matoza, 2013) as well as the plumbing system pressurization suggested by the GPS areal strain. This pressurization phenomenon eventually led to a dike intrusion accompanied by a seismic swarm and culminated with the development of erupting fissures at the eastern base of the NSEC on 24 December 2018 (Figure 1). The spatial distribution of dv/v indicates that there is a correlation between the highest drop and where the eruption took place (Figure 3). The seismicity mostly took place at a shallower depth (0–6 km) (Alparone et al., 2020) and evolved into the  $M_r$ 4.8 earthquake on 26 December along the strike-slip FF (Figure 1b). Indeed, the dike intrusion along the flank also caused the acceleration of the large-scale seaward sliding of the eastern flank, which likely contributed to the occurrence of the  $M_1$  4.8 earthquake (Aloisi et al., 2020). This flank sliding occurred at rates significantly larger than past slip events (Mattia et al., 2015) and suddenly depressurized the dike intrusion at the shallow depth. This likely remobilized the magma residing at the shallow depth and contributed to ending the 4-day eruption, which otherwise had strong indicators for an event much larger in terms of erupted volumes. Eventually, dv/vstarts recovering and displays an abrupt increase 0.2% after this phase. This subsequent increase of dv/v could be due to the closing of the cracks as the volcano shows the deflation behavior. However, it should be noted that the error of dv/v is high ( $\pm 0.3\%$ ) in this period. This is likely due to the changes that took place in the medium during the eruption, which led to sharp modifications of the CCF compared to the reference CCF. Consequently, we do not rely on the post-eruption dv/v data since the individual station pairs show scattered seismic velocity changes (Figure S9 in Supporting Information S1) and CCFs become unstable after the eruption.

# 5. Conclusions

We implemented a seismic ambient noise interferometry technique for the 2018 Mt. Etna flank eruption by calculating the cross-correlations of the continuous signal recorded by a dense network. We used the moving-window cross-spectral analysis on phase-weighted daily stacked vertical components to successfully identify seismic velocity changes in the pre-eruption period mostly localized around the summit and flank areas. The observed temporal variation of seismic velocities (dv/v) is associated with the pressurization of the plumbing system within the volcano edifice during the period preceding the eruption. We integrated dv/v results with VTs, volcanic tremor, volcanic degassing, gravity, and ground deformation to provide more perspective on how it relates to the dynamics of the magma ascent.

Combining all these observables confirmed the relationship between the magma intrusion and the observed reduction in seismic velocities and provided an additional perspective on the impact of two significant earthquakes on the development of magma transport and how they related to the changes in the local stress field and ground deformation.

Ultimately, using dv/v helped provide a better context to understand the lead-up to an eruption that had the potential of becoming much larger but was aborted after the stress release that followed the shallow earthquake caused the intrusion to stop.



# **Data Availability Statement**

The seismic ambient noise analysis in this study was applied via MSNoise python package (Lecocq et al., 2014). The maps were prepared using the GMT software (Wessel et al. (2019), Python (http://www.python.org), and MatPlotLib (https://matplotlib.org/) visualization packages. Seismic data used in the study are available through FDSN (INGV Seismological Data Centre, 1997). The seismic catalog was obtained by Alparone et al. (2020). For topography, SRTM-1 data were used (Farr et al., 2007). The probabilistic source inversion method was performed with the Grond framework (Heimann et al., 2018). The rainfall data are taken from "SIAS Servizio Informativo Agrometeorologico Siciliano" (http://www.sias.regione.sicilia.it/).

### References

- Aloisi, M., Bonaccorso, A., Cannavò, F., Currenti, G., & Gambino, S. (2020). The 24 December 2018 eruptive intrusion at Etna volcano as revealed by multidisciplinary continuous deformation networks (CGPS, borehole strainmeters and tiltmeters). Journal of Geophysical Research: Solid Earth, 125(8), 2002–2003. https://doi.org/10.1029/2019JB019117
- Aloisi, M., Cocina, O., Neri, G., Orecchio, B., & Privitera, E. (2002). Seismic tomography of the crust underneath the Etna volcano, Sicily. *Physics of the Earth and Planetary Interiors*, 134(3), 139–155. https://doi.org/10.1016/S0031-9201(02)00153-X
- Alparone, S., Barberi, G., Cocina, O., Giampiccolo, E., Musumeci, C., & Patanè, D. (2012). Intrusive mechanism of the 2008 2009 Mt. Etna eruption: Constraints by tomographic images and stress tensor analysis. *Journal of Volcanology and Geothermal Research*, 229–230, 50–63. https://doi.org/10.1016/j.jvolgeores.2012.04.001
- Alparone, S., Barberi, G., Giampiccolo, E., Maiolino, V., Mostaccio, A., Musumeci, C., et al. (2020). Seismological constraints on the 2018 Mt. Etna (Italy) flank eruption and implications for the flank dynamics of the volcano. *Terra Nova*, *32*(5), 334–344. https://doi.org/10.1111/ ter.12463
- Alparone, S., Maiolino, V., Mostaccio, A., Scaltrito, A., Ursino, A., Barberi, G., & Zuc- carello, L. (2015). Mt. Etna seismic catalog 2000-2010. https://doi.org/10.13127/ETNASC/2000\_2010
- Azzaro, R., Branca, S., Gwinner, K., & Coltelli, M. (2012). The volcano-tectonic map of Etna volcano, 1:100.000 scale: An integrated approach based on a morphotectonic analysis from high-resolution DEM constrained by geologic, active faulting and seismotectonic data. *Italian Jour*nal of Geosciences, 131(1), 153–170. https://doi.org/10.3301/IJG.2011.29
- Behncke, B., Branca, S., Corsaro, R. A., De Beni, E., Miraglia, L., & Proietti, C. (2014). The 2011–2012 summit activity of Mount Etna: Birth, growth and products of the new SE crater. Journal of Volcanology and Geothermal Research, 270, 10–21.
- Bennington, N. L., Haney, M., De Angelis, S., Thurber, C. H., & Freymueller, J. (2015). Monitoring changes in seismic velocity related to an ongoing rapid inflation event at Okmok volcano, Alaska. *Journal of Geophysical Research: Solid Earth*, 120(8), 5664–5676. https://doi. org/10.1002/2015JB011939
- Bensen, G. D., Ritzwoller, M. H., Barmin, M. P., Levshin, A. L., Lin, F., Moschetti, M. P., & Yang, Y. (2007). Processing seismic ambient noise data to obtain reliable broad-band surface wave dispersion measurements. *Geophysical Journal International*, 169(3), 1239–1260.
- Bonaccorso, A., Cianetti, S., Giunchi, C., Trasatti, E., Bonafede, M., & Boschi, E. (2005). Analytical and 3-D numerical modelling of Mt. Etna (Italy) volcano inflation. *Geophysical Journal International*, 163(2), 852–862. https://doi.org/10.1111/j.1365-246X.2005.02777.x
- Bonforte, A., Branca, S., & Palano, M. (2007). Geometric and kinematic variations along the active Pernicana fault: Implication for the dynamics of Mount Etna NE flank (Italy). Journal of Volcanology and Geothermal Research, 160(1–2), 210–222. https://doi.org/10.1016/ jvolgeores.2006.08.009
- Bonforte, A., Guglielmino, F., & Puglisi, G. (2019). Large dyke intrusion and small eruption: The December 24, 2018 Mt. Etna eruption imaged by Sentinel-1 data. *Terra Nova*, 31(4), 405–412. https://doi.org/10.1111/ter.12403
- Borzi, A. M., Giuffrida, M., Zuccarello, F., Palano, M., & Viccaro, M. (2020). The Christmas 2018 eruption at Mount Etna: Enlightening how the volcano factory works through a multiparametric inspection. *Geochemistry, Geophysics, Geosystems*, 21(10), 1–15. https://doi. org/10.1029/2020GC009226
- Branca, S., Coltelli, M., Groppelli, G., & Lentini, F. (2011). Geological map of Etna volcano, 1:50,000 scale. *Italian Journal of Geosciences*, 130(3), 265–291. https://doi.org/10.3301/IJG.2011.15
- Brenguier, F., Campillo, M., Takeda, T., Aoki, Y., Shapiro, N. M., Briand, X., & Miyake, H. (2014). Mapping pressurized volcanic fluids from induced crustal seismic velocity drops. *Science*, 345(6192), 80–82. https://doi.org/10.1126/science.1254073
- Brenguier, F., Shapiro, N. M., Campillo, M., Ferrazzini, V., Duputel, Z., Coutant, O., & Nercessian, A. (2008). Towards forecasting volcanic eruptions using seismic noise. *Nature Geoscience*, 1(2), 126–130. https://doi.org/10.1038/ngeo104
- Büyükakpınar, P., Cesca, S., Hainzl, S., Jamalreyhani, M., Heimann, S., & Dahm, T. (2021). Reservoir-triggered earthquakes around the Atatürk Dam (Southeastern Turkey). Frontiers of Earth Science, 9, 353. https://doi.org/10.3389/feart.2021.663385
- Cannata, A. (2012). Crustal changes at Mt. Etna volcano accompanying the 2002–2003 eruption as inferred from a repeating earthquake analysis. *Geophysical Research Letters*, 39(18). https://doi.org/10.1029/2012GL053185
- Cannata, A., Cannavò, F., Montalto, P., Ercoli, M., Mancinelli, P., Pauselli, C., & Leto, G. (2017). Monitoring crustal changes at volcanoes by seismic noise interferometry: Mt. Etna case of study. *Journal of Volcanology and Geothermal Research*, 337, 165–174. https://doi.org/10.1016/j. jvolgeores.2017.03.023
- Cannata, A., Di Grazia, G., Giuffrida, M., Gresta, S., Palano, M., Sciotto, M., et al. (2018). Space-time evolution of magma storage and transfer at Mt. Etna volcano (Italy): The 2015–2016 reawakening of Voragine crater. *Geochemistry, Geophysics, Geosystems*, 19(2), 471–495. https:// doi.org/10.1002/2017GC007296
- Cannata, A., Spedalieri, G., Behncke, B., Cannavò, F., Di Grazia, G., Gambino, S., & Palano, M. (2015). Pressurization and depressurization phases inside the plumbing sys-tem of Mount Etna volcano: Evidence from a multiparametric approach. *Journal of Geophysical Research: Solid Earth*, 120(9), 5965–5982. https://doi.org/10.1002/2015JB012227
- Cannavò, F. (2019). A new user-friendly tool for rapid modelling of ground deformation. Computers & Geosciences, 128, 60–69.
- Cannavò, F., Sciotto, M., Cannata, A., & Di Grazia, G. (2019). An integrated geophysical approach to track magma intrusion: The 2018 Christmas Eve eruption at mount Etna. *Geophysical Research Letters*, *46*(14), 8009–8017. https://doi.org/10.1029/2019GL083120
- Carbone, D., Cannavò, F., Greco, F., Reineman, R., & Warburton, R. J. (2019). The benefits of using a network of superconducting gravimeters to monitor and study active volcanoes. *Journal of Geophysical Research: Solid Earth*, 124(4), 4035–4050. https://doi.org/10.1029/2018JB017204

# Acknowledgments The research has benefited from funding

provided by the Italian Presidenza del Consiglio dei Ministri-Dipartimento della Protezione Civile and Istituto Nazionale di Geofisica e Vulcanologia, DPC-INGV Allegato B2 2019-2021-Task 10 "TRUST" (grant number: OB.FU.9996.030) (Scientific Responsibility: F.C.). This paper does not necessarily represent DPC's official opinions and policies. P.B. has also received funding by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) Project 407141557. A.C. thanks CHANCE project, II Edition, Università degli Studi di Catania (principal investigator A. Cannata) and grant PIACERI, 2020-2022 programme (PAROSSISMA project, code 22722132140; principal investigator Marco Viccaro). The study was partially funded by the H2020 NEWTON-g project (Grant Agreement 801221). Open Access Funding provided by Istituto Nazionale di Geofisica e Vulcanologia within the CRUI-CARE Agreement. We are very grateful to Mehdi Nikkhoo for his constructive discussion and comments on this work. We are indebted to the technicians and technologists of the INGV Osservatorio Etneo and Sezione di Palermo for enabling and improving the acquisition of data. We also thank the editor and the reviewers for their valuable comments and suggestions that helped us to improve the article.

- Carbone, D., Poland, M. P., Diament, M., & Greco, F. (2017). The added value of time-variable microgravimetry to the understanding of how volcanoes work. *Earth-Science Reviews*, 169, 146–179. https://doi.org/10.1016/j.earscirev.2017.04.014
- Chauhan, M. S., Cannavó, F., Carbone, D., & Greco, F. (2020). Insights into Mount Etna December 2018 eruption from joint inversion of deformation and gravity data. *Geophysical Research Letters*, 47(16), e2020GL087786. https://doi.org/10.1029/2020GL087786
- Chouet, B. A., & Matoza, R. S. (2013). A multi-decadal view of seismic methods for detecting precursors of magma movement and eruption. Journal of Volcanology and Geothermal Research, 252, 108–175. https://doi.org/10.1016/j.jvolgeores.2012.11.013
- Civico, R., Pucci, S., Nappi, R., Azzaro, R., Villani, F., Pantosti, D., et al. (2019). Surface ruptures following the 26 December 2018, M<sub>w</sub> 4.9, Mt. Etna earthquake, Sicily (Italy). *Journal of Maps*, *15*(2), 831–837.
- Clarke, D., Brenguier, F., Froger, J.-L., Shapiro, N. M., Peltier, A., & Staudacher, T. (2013). Timing of a large volcanic flank movement at Piton de la Fournaise Volcano using noise-based seismic monitoring and ground deformation measurements. *Geophysical Journal International*, 195(2), 1132–1140. https://doi.org/10.1093/gji/ggt276
- Crafford, A., & Venzke, E. (2018). Global volcanism program, 2018. Report on Etna (Italy). Bulletin of the Global Volcanism Network, 43. Smithsonian Institution. https://doi.org/10.5479/si.GVP.BGVN201812-211060
- Cubuk-Sabuncu, Y., Jónsdóttir, K., Caudron, C., Lecocq, T., Parks, M. M., Geirsson, H., & Mordret, A. (2021). Temporal seismic velocity changes during the 2020 rapid inflation at Mt. Porbjörn-Svartsengi, Iceland, using seismic ambient noise. *Geophysical Research Letters*, 48(11), e2020GL092265. https://doi.org/10.1029/2020GL092265
- De Beni, E., Behncke, B., Branca, S., Nicolosi, I., Carluccio, R., D'Ajello Caracciolo, F., & Chiappini, M. (2015). The continuing story of Etna's New Southeast Crater (2012- 2014): Evolution and volume calculations based on field surveys and aerophotogrammetry. *Journal of Volcanology and Geothermal Research*, 303(September), 175–186. https://doi.org/10.1016/j.jvolgeores.2015.07.021
- De Novellis, V., Atzori, S., De Luca, C., Manzo, M., Valerio, E., Bonano, M., et al. (2019). DInSAR analysis and analytical modeling of Mount Etna Displacements: The December 2018 volcano-tectonic crisis. *Geophysical Research Letters*, 46(11), 5817–5827. https://doi. org/10.1029/2019GL082467
- De Plaen, R. S. M., Cannata, A., Cannavo', F., Caudron, C., Lecocq, T., & Francis, O. (2019). Temporal changes of seismic velocity caused by volcanic activity at Mt. Etna revealed by the Autocorrelation of ambient seismic noise. *Frontiers of Earth Science*, 6(January), 1–11. https:// doi.org/10.3389/feart.2018.00251
- De Plaen, R. S. M., Lecocq, T., Caudron, C., Ferrazzini, V., & Francis, O. (2016). Single-station monitoring of volcanoes using seismic ambient noise. Geophysical Research Letters, 43(16), 8511–8518. https://doi.org/10.1002/2016GL070078
- Donaldson, C., Caudron, C., Green, R. G., Thelen, W. A., & White, R. S. (2017). Relative seismic velocity variations correlate with deformation at Kīlauea volcano. Science Advances, 3(6), e1700219. https://doi.org/10.1126/sciadv.1700219
- Duputel, Z., Ferrazzini, V., Brenguier, F., Shapiro, N., Campillo, M., & Nercessian, A. (2009). Real time monitoring of relative velocity changes using ambient seismic noise at the Piton de la Fournaise volcano (La Réunion) from January 2006 to June 2007. Journal of Volcanology and Geothermal Research, 184(1), 164–173. https://doi.org/10.1016/j.jvolgeores.2008.11.024
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., et al. (2007). The shuttle radar topography mission. *Reviews of Geophysics*, 45(2). https://doi.org/10.1029/2005RG000183
- Gambino, S., Cannata, A., Cannavò, F., La Spina, A., Palano, M., Sciotto, M., et al. (2016). The unusual 28 December 2014 dike-fed paroxysm at Mount Etna: Timing and mechanism from a multidisciplinary perspective. *Journal of Geophysical Research: Solid Earth*, 121(3), 2037–2053. https://doi.org/10.1002/2015JB012379
- Gurrieri, S., Liuzzo, M., Giuffrida, G., & Boudoire, G. (2021). The first observations of CO<sub>2</sub> and CO<sub>2</sub>/SO<sub>2</sub> degassing variations recorded at Mt. Etna during the 2018 eruptions followed by three strong earthquakes. *Italian Journal of Geosciences*, *140*(1), 1–12. https://doi.org/10.3301/ IJG.2020.25
- Heimann, S., Isken, M., Kühn, D., Sudhaus, H., Steinberg, A., Vasyura-Bathke, H., et al. (2018). Grond A probabilistic earthquake source inversion framework. https://doi.org/10.5880/GFZ.2.1.2018.003
- Heimann, S., Vasyura-Bathke, H., Sudhaus, H., Isken, M. P., Kriegerowski, M., Steinberg, A., & Dahm, T. (2019). A Python framework for efficient use of pre-computed Green's functions in seismological and other physical forward and inverse source problems. *Solid Earth*, 10(6), 1921–1935. https://doi.org/10.5194/se-10-1921-2019
- Herring, T., King, R. W., Floyd, M., & McClusky, S. (2010). Introduction to GAMIT/GLOBK introduction to GAMIT/GLOBK.
- Herrmann, R. B. (2013). Computer programs in seismology: An evolving tool for instruction and research. *Seismological Research Letters*, 84, 1081–1088.
- Hill, D., & Prejean, S. (2015). Dynamic triggering. In Treatise on geophysics (pp. 273-304). Elsevier. https://doi.org/10.1016/b978-0-444-53802-4.00078-6
- Hotovec-Ellis, A. J., Vidale, J. E., Gomberg, J., Thelen, W., & Moran, S. C. (2015). Changes in seismic velocity during the first 14 months of the 2004–2008 eruption of Mount St. Helens, Washington. *Journal of Geophysical Research: Solid Earth*, 120(9), 6226–6240. https://doi. org/10.1002/2015JB012101
- Hutapea, F. L., Tsuji, T., & Ikeda, T. (2020). Real-time crustal monitoring system of Japanese Islands based on spatio-temporal seismic velocity variation. *Earth Planets and Space*, 72(1), 19. https://doi.org/10.1186/s40623-020-1147-y

INGV Seismological Data Centre. (1997). Rete Sismica Nazionale (RSN). Istituto Nazionale di Geofisica e Vulcanologia (INGV). https://doi. org/10.13127/SD/X0FXNH7QFY

- Jamalreyhani, M., Pousse-Beltran, L., Büyükakpınar, P., Cesca, S., Nissen, E., Gh- ods, A., et al. (2021). The 2019–2020 Khalili (Iran) Earthquake sequence—Anthropogenic seismicity in the Zagros simply folded Belt? *Journal of Geophysical Research: Solid Earth*, 126(12), e2021JB022797. https://doi.org/10.1029/2021JB022797
- Kennett, B. L. N., Engdahl, E. R., & Buland, R. (1995). Constraints on seismic velocities in the Earth from traveltimes. *Geophysical Journal International*, 122(1), 108–124. https://doi.org/10.1111/j.1365-246X.1995.tb03540.x
- Kühn, D., Heimann, S., Isken, M. P., Ruigrok, E., & Dost, B. (2020). Probabilistic moment tensor inversion for hydrocarbon-induced seismicity in the Groningen gas field, The Netherlands, Part 1: Testing. *Bulletin of the Seismological Society of America*, 110(5), 2095–2111. https://doi. org/10.1785/0120200099
- Laiolo, M., Ripepe, M., Cigolini, C., Coppola, D., Della Schiava, M., Genco, R., & Silengo, M. C. (2019). Space- and ground-based geophysical data tracking of magma migration in shallow feeding system of Mount Etna volcano. *Remote Sensing*, 11(10). https://doi.org/10.3390/ rs11101182
- Lecocq, T., Caudron, C., & Brenguier, F. (2014). MSNoise, a Python package for monitoring seismic velocity changes using ambient seismic noise. Seismological Research Letters, 85(3), 715–726. https://doi.org/10.1785/0220130073

- Lecocq, T., Longuevergne, L., Pedersen, H. A., Brenguier, F., & Stammler, K. (2017). Monitoring ground water storage at mesoscale using seismic noise: 30 years of continuous observation and thermo-elastic and hydrological modeling. *Scientific Reports*, 7(1), 14241. https://doi. org/10.1038/s41598-017-14468-9
- Liuzzo, M., Gurrieri, S., Giudice, G., & Giuffrida, G. (2013). Ten years of soil CO2 continuous monitoring on Mt. Etna: Exploring the relationship between processes of soil degassing and volcanic activity. *Geochemistry, Geophysics, Geosystems*, 14(8), 2886–2899. https://doi. org/10.1002/ggge.20196
- Loughlin, S., Vye-Brown, C., Sparks, R., Brown, S., Barclay, J., Calder, E., & Valentine, G. (2015). An introduction to global volcanic hazard and risk. In *Global volcanic hazards and risk* (pp. 1–79). Cambridge University Press. https://doi.org/10.1017/CBO9781316276273.003
- Mattia, M., Bruno, V., Caltabiano, T., Cannata, A., Cannavò, F., D'Alessandro, W., et al. (2015). A comprehensive interpretative model of slow slip events on Mt. Etna's eastern flank. *Geochemistry, Geophysics, Geosystems*, 16(3), 635–658. https://doi.org/10.1002/2014GC005585
- Mattia, M., Bruno, V., Montgomery-Brown, E., Patanè, D., Barberi, G., & Coltelli, M. (2020). Combined seismic and geodetic analysis before, during, and after the 2018 Mount Etna eruption. *Geochemistry, Geophysics, Geosystems*, 21(9), e2020GC009218. https://doi.org/ 10.1029/2020GC009218
- Monaco, C., Catalano, S., Cocina, O., De Guidi, G., Ferlito, C., Gresta, S., et al. (2005). Tectonic control on the eruptive dynamics at Mt. Etna Volcano (Sicily) during the 2001 and 2002-2003 eruptions. *Journal of Volcanology and Geothermal Research*, *144*(1–4), 211–233. https://doi.org/10.1016/j.jvolgeores.2004.11.024
- Neri, M., Maio, M. D., Crepaldi, S., Suozzi, E., Lavy, M., Marchionatti, F., & Buongiorno, M. F. (2017). Topographic maps of Mount Etna's summit craters, updated to December 2015. *Journal of Maps*, 13(2), 674–683. https://doi.org/10.1080/17445647.2017.1352041
- Obermann, A., Planès, T., Larose, E., & Campillo, M. (2013). Imaging preeruptive and coeruptive structural and mechanical changes of a volcano with ambient seismic noise. *Journal of Geophysical Research: Solid Earth*, 118(12), 6285–6294. https://doi.org/10.1002/2013JB010399
- Olivier, G., Brenguier, F., Carey, R., Okubo, P., & Donaldson, C. (2019). Decrease in seismic velocity observed prior to the 2018 eruption of Kīlauea volcano with ambi-ent seismic noise interferometry. *Geophysical Research Letters*, 46(7), 3734–3744. https://doi.org/10.1029/2018GL081609 Palano, M., Rossi, M., Cannavò, F., Bruno, V., Aloisi, M., Pellegrino, D., et al. (2010). Etn@ ref: A geodetic reference frame for Mt. Etna GPS
- networks. Annals of Geophysics, 53(4), 49–57.
  Paonita, A., Liuzzo, M., Salerno, G., Federico, C., Bonfanti, P., Caracausi, A., & Giudice, G. (2021). Intense overpressurization at basaltic open-conduit volcanoes as inferred by geochemical signals: The case of the Mt. Etna December 2018 eruption. Science Advances, 7(36). https://doi.org/10.1126/sciady.abg6297
- Patanè, D., Aiuppa, A., Aloisi, M., Behncke, B., Cannata, A., Coltelli, M., & Salerno, G. (2013). Insights into magma and fluid transfer at Mount Etna by a multiparametric approach: A model of the events leading to the 2011 eruptive cycle. *Journal of Geophysical Research: Solid Earth*, 118(7), 3519–3539. https://doi.org/10.1002/jgrb.50248
- Patanè, D., Gori, P. D., Chiarabba, C., & Bonaccorso, A. (2003). Magma ascent and the pressurization of Mount Etna's volcanic system. *Science*, 299(5615), 2061–2063. https://doi.org/10.1126/science.1080653
- Petersen, G. M., Cesca, S., Heimann, S., Niemz, P., Dahm, T., & Kühn, D. (2021). Regional centroid moment tensor inversion of small to moderate earthquakes in the Alps using the dense AlpArray seismic network: Challenges and seismotectonic insights. *Solid Earth*, 12(6), 1233–1257. https://doi.org/10.5194/se-12-1233-2021
- Pondrelli, S. (2002). European-Mediterranean Regional Centroid-Moment Tensors Catalog (RCMT). Istituto Nazionale di Geofisica e Vulcanologia (INGV). https://doi.org/10.13127/RCMT/EUROMED
- Ratdomopurbo, A., & Poupinet, G. (1995). Monitoring a temporal change of seismic velocity in a volcano: Application to the 1992 eruption of Mt. Merapi (Indonesia). *Geophysical Research Letters*, 22(7), 775–778. https://doi.org/10.1029/95GL00302
- Ruiz, M. Z., Civilini, F., Ebinger, C. J., Oliva, S. J., Ruiz, M. C., Badi, G., & Casas, J. A. (2022). Precursory signal Detected for the 2018 Sierra Negra volcanic eruption, Galápagos, using seismic ambient noise. *Journal of Geophysical Research: Solid Earth*, 127(3), e2021JB022990. https://doi.org/10.1029/2021JB022990
- Rust, D., & Neri, M. (1996). The boundaries of large-scale collapse on the flanks of Mount Etna, Sicily. *Geological Society Special Publications*, 110(January), 193–208. https://doi.org/10.1144/GSL.SP.1996.110.01.15
- Salerno, G. G., Burton, M. R., Oppenheimer, C., Caltabiano, T., Randazzo, D., Bruno, N., & Longo, V. (2009). Three-years of SO<sub>2</sub> flux measurements of Mt. Etna using an automated UV scanner array: Comparison with conventional traverses and uncertainties in flux retrieval. *Journal* of Volcanology and Geothermal Research, 183, 76–83.
- Scarfi, L., Messina, A., & Cassisi, C. (2013). Sicily and southern Calabria focal mechanism database: A valuable tool for local and regional stress-field determination. *Annals of Geophysics*, 56(1), 0109.
- Schimmel, M., & Paulssen, H. (1997). Noise reduction and detection of weak, coherent signals through phase weighted stacks. *Geophysical Journal International*, 130, 497–505. https://doi.org/10.1111/j.1365-246X.1997.tb05664.x
- Scognamiglio, L., Tinti, E., & Quintiliani, M. (2006). *Time Domain Moment Tensor (TDMT)*. Istituto Nazionale di Geofisica e Vulcanologia (INGV). https://doi.org/10.13127/TDMT
- Sens-Schönfelder, C., & Wegler, U. (2006). Passive image interferemetry and seasonal variations of seismic velocities at Merapi Volcano, Indonesia. Geophysical Research Letters, 33(21), 1–5. https://doi.org/10.1029/2006GL027797
- Spina, L., Taddeucci, J., Cannata, A., Sciotto, M., Bello, E. D., Scarlato, P., & Dingwell, D. B. (2017). Time-series analysis of fissure-fed multi-vent activity: A snapshot from the July 2014 eruption of Etna volcano (Italy). *Bulletin of Volcanology*, 79(7). https://doi.org/10.1007/ s00445-017-1132-x
- Taira, T., & Brenguier, F. (2016). Response of hydrothermal system to stress transients at Lassen Volcanic Center, California, inferred from seismic interferometry with ambient noise. *Earth Planets and Space*, 68(162). https://doi.org/10.1186/s40623-016-0538-6
- Van Camp, M., de Viron, O., Watlet, A., Meurers, B., Francis, O., & Caudron, C. (2017). Geophysics from terrestrial time-variable gravity measurements. *Reviews of Geophysics*, 55(4), 938–992. https://doi.org/10.1002/2017RG000566
- Viccaro, M., Giuffrida, M., Zuccarello, F., Scandura, M., Palano, M., & Gresta, S. (2019). Violent paroxysmal activity drives self-feeding magma replenishment at Mt. Etna. *Scientific Reports*, 9(1), 1–14. https://doi.org/10.1038/s41598-019-43211-9
- Viccaro, M., Zuccarello, F., Cannata, A., Palano, M., & Gresta, S. (2016). How a complex basaltic volcanic system works: Constraints from integrating seismic, geodetic, and petrological data at Mount Etna volcano during the July–August 2014 eruption. *Journal of Geophysical Research: Solid Earth*, 121(8), 5659–5678. https://doi.org/10.1002/2016JB013164
- Villani, F., Pucci, S., Azzaro, R., Civico, R., Cinti, F. R., Pizzimenti, L., et al. (2020). Surface ruptures database related to the 26 December 2018, MW 4.9 Mt. Etna earthquake, southern Italy. *Scientific Data*, 7(1), 1–9. https://doi.org/10.1038/s41597-020-0383-0
- Wessel, P., Luis, J. F., Uieda, L., Scharroo, R., Wobbe, F., Smith, W. H. F., & Tian, D. (2019). The generic mapping tools version 6. Geochemistry, Geophysics, Geosystems, 20(11), 5556–5564. https://doi.org/10.1029/2019GC008515



- Yang, X.-M., Davis, P. M., & Dieterich, J. H. (1988). Deformation from inflation of a dipping finite prolate spheroid in an elastic half-space as a model for volcanic stressing. *Journal of Geophysical Research*, 93(B5), 4249–4257.
- Zhan, Z., Tsai, V. C., & Clayton, R. W. (2013). Spurious velocity changes caused by temporal variations in ambient noise frequency content. *Geophysical Journal International*, 194(3), 1574–1581. https://doi.org/10.1093/gji/ggt170
- Zhang, H., Thurber, C., & Bedrosian, P. (2009). Joint inversion for Vp, Vs, and Vp/Vs at SAFOD, Parkfield, California. *Geochemistry, Geophysics, Geosystems*, 10(11). https://doi.org/10.1029/2009GC002709